Author's response to reviews

Title: Rhinosinusitis in morbidity registrations in Dutch General Practice: A retrospective case-control study

Authors:

   Ruth Hoffmans (r.hoffmans@amc.nl)
   Tjard Schermer (Tjard.Schermer@radboudumc.nl)
   Karin van der Linde (kjesselink@hotmail.com)
   Kees van Boven (cvboven@hetnet.nl)
   Chris van Weel (Chris.vanWeel@radboudumc.nl)
   Wytske Fokkens (w.j.fokkens@amc.nl)

Version: 3
Date: 11 August 2015

Author's response to reviews: see over
Dear Dr. Tsiligiani,

With reference to your letter with editorial revision instructions of 7th of August 2015, we would like to thank you for reviewing our revised manuscript entitled “Rhinocuronitis in morbidity registrations in Dutch General Practice: A retrospective case-control study”. We are happy to hear that our manuscript is suitable for publication in the journal.

The following editorial revisions are made:
- The title page was provided with the email addresses of all authors
- The introduction was renamed as Background
- In the methods section we confirm that permission was granted to access the used databases
- The section Funding was removed
- Author Contributorship was renamed as Author’s Contributions

We would like to submit our revised manuscript and we look forward to the publication in BMC Family Practice.

If you have any further questions, please do not hesitate to contact us.

On behalf of my co-authors,

Sincerely,

Ruth Hoffmans